COVID-19 Vaccine Safety



COVID 19-vaccines are safe and effective.

They can keep you from getting and spreading the virus that causes COVID-19. COVID-19 vaccines also help keep you from getting seriously ill even if you do get COVID-19. Getting vaccinated yourself may also protect people around you, particularly people at increased risk for severe illness from COVID-19.

Over 369 million doses of COVID-19 vaccine have been given in the United States from December 14, 2020, through August 30, 2021. *COVID-19 vaccines are safe* and effective. COVID-19 vaccines were evaluated in tens of thousands of participants in clinical trials. The vaccines met the Food and Drug Administration's (FDA) rigorous scientific standards for safety, effectiveness, and manufacturing quality needed to support approval or authorization of a vaccine.

- COVID-19 vaccines were developed using science that has been around for decades.
- COVID-19 vaccines are not experimental. They went through all the required stages of clinical trials. Extensive testing and monitoring have shown that these vaccines are safe and effective.
- COVID-19 vaccines have received and continue to undergo the most intensive safety monitoring in U.S. history.

Adverse events and serious health problems following COVID-19 vaccination are rare. To date, the systems in place to monitor the safety of these vaccines have found only two serious types of health problems after vaccination, both of which are rare.

- A small number of people have had a severe allergic reaction (called "anaphylaxis") after vaccination, but this is rare occurring in approximately 2 to 5 people per million vaccinated in the U.S. Anaphylaxis can occur after any vaccination. If this occurs, vaccination providers have medicines available to effectively and immediately treat the reaction. After you get a COVID-19 vaccine, you will be asked to stay for 15–30 minutes so you can be observed in case you have a severe allergic reaction and need immediate treatment.
- After receiving the J&J/Janssen COVID-19 Vaccine, there is risk for a rare but serious adverse event—blood clots with low platelets (thrombosis with thrombocytopenia syndrome, or TTS). Women younger than 50 years old should especially be aware of their increased risk for this rare adverse event. There are other COVID-19 vaccines available for which this risk has not been seen. This adverse event is rare, occurring at a rate of about 7 per 1 million vaccinated women between 18 and 49 years old (as of August 25, 2021 the CDC has identified 44 confirmed reports out of 14.2 million doses of the J&J vaccine). For women 50 years and older and men of all ages, this adverse event is even more rare.
- CDC and FDA are monitoring reports of Guillain-Barré Syndrome (GBS) in people who have received the J&J/Janssen COVID-19 Vaccine. GBS is a rare disorder where the body's immune system damages nerve cells, causing muscle weakness and sometimes paralysis. Most people fully recover from GBS, but some have permanent nerve damage. After more than 14.2 million J&J/Janssen COVID-19 Vaccine doses administered, there have been around 176 preliminary reports of GBS identified. These cases have largely been reported about 2 weeks after vaccination and mostly in men, many 50 years and older.
- Myocarditis and pericarditis after COVID-19 vaccination are rare. As of August 25, 2021, VAERS has received 1,377 reports of
 myocarditis or pericarditis among people ages 30 and younger who received COVID-19 vaccine. Most cases have been reported after
 mRNA COVID-19 vaccination (Pfizer-BioNTech or Moderna), particularly in male adolescents and young adults. Through follow-up,
 including medical record reviews, CDC and FDA have confirmed 798 reports of myocarditis or pericarditis. CDC and its partners are
 investigating these reports to assess whether there is a relationship to COVID-19 vaccination.
- Reports of death after COVID-19 vaccination are rare. More than 369 million doses of COVID-19 vaccines were administered in the United States from December 14, 2020, through August 30, 2021. During this time, VAERS received 7,218 reports of death (0.0020%) among people who received a COVID-19 vaccine. FDA requires healthcare providers to report any death after COVID-19 vaccination to VAERS, even if it's unclear whether the vaccine was the cause. Reports of adverse events to VAERS following vaccination, including deaths, do not necessarily mean that a vaccine caused a health problem. A review of available clinical information, including death certificates, autopsy, and medical records, has not established a causal link to COVID-19 vaccines. However, recent reports indicate a plausible causal relationship between the J&J/Janssen COVID-19 Vaccine and TTS, a rare and serious adverse event—blood clots with low platelets—which has caused deaths.

RESULTS FROM VACCINE SAFETY MONITORING EFFORTS ARE REASSURING. SOME PEOPLE HAVE NO SIDE EFFECTS. OTHERS HAVE REPORTED COMMON SIDE EFFECTS AFTER COVID-19 VACCINATION, LIKE:

- + SWELLING, REDNESS, AND PAIN AT INJECTION SITE
- + FEVER
- + HEADACHE
- + TIREDNESS
- + MUSCLE PAIN
- + CHILLS



THE FEDERAL GOVERNMENT IS PROVIDING THE VACCINE free of charge to all people living in the United STATES, REGARDLESS OF THEIR IMMIGRATION OR HEALTH INSURANCE STATUS. THERE ARE SEVERAL WAYS YOU CAN LOOK FOR VACCINATION PROVIDERS NEAR YOU IN THE UNITED STATES.

Visit <u>Vaccines.gov</u> to find vaccination providers near you.

Text your ZIP code to 438829 or call 1-800-232-0233 to find vaccine locations near you in the United States.

Check your local pharmacy's website to see if vaccination appointments are available. Find out which pharmacies are participating in the Federal Retail Pharmacy Program.

Contact your state health department to find additional vaccination locations in the area.

Check your local news outlets. They may have information on how to get a vaccination appointment.

The CDC continues to recommend the use of the newly FDA-approved Pfizer-BioNTech COVID-19 Vaccine for people 16 years and older, as one of the recommended vaccines to protect against COVID-19.

COVID-19 booster shots may be available beginning in the fall, with individuals being eligible starting approximately 8 months after they received their final dose of vaccine. Individuals will need to receive the same vaccine brand as their original COVID-19 vaccine. This is subject to authorization by the FDA and recommendation by CDC's Advisory Committee on Immunization Practices (ACIP).

Vaccine Brand Name	Who Can Get this Vaccine ¹	How Many Shots You Will Need	When Are You Fully Vaccinated?
Pfizer-BioNTech	People 12 years and older	2 shots	2 weeks after your second
		Given 3 weeks (21 days) apart ²	shot
Moderna	People 18 years and older	2 shots	2 weeks after your second
		Given 4 weeks (28 days) apart ²	shot
Johnson & Johnson's	People 18 years and older	1 shot	2 weeks after your shot
Janssen			
¹ If you have had a severe allergic reaction (anaphylaxis) or an immediate allergic reaction to any ingredient in the vaccine you are scheduled to receive, you should not get that vaccine. If you have been instructed not to get one type of COVID-19 vaccine, you may still be able to get another type. Learn more information for people with allergies.			
² You should get your second shot as close to the recommended 3-week or 4-week interval as possible. However, your second shot may be given up to 6 weeks (42 days) after the first dose, if necessary.			

- All COVID-19 vaccines currently authorized in the United States are effective against COVID-19, including serious outcomes of severe disease, hospitalization, and death.
- Available evidence suggests the currently authorized mRNA COVID-19 vaccines (Pfizer-BioNTech and Moderna) are highly effective against hospitalization and death for a variety of strains, including Alpha (B.1.1.7), Beta (B.1.351), Gamma (P.1), and Delta (B.1.617.2); data suggest lower effectiveness against confirmed infection and symptomatic disease caused by the Beta, Gamma, and Delta variants compared with the ancestral strain and Alpha variant. Ongoing monitoring of vaccine effectiveness against variants is needed.
- A growing body of evidence indicates that people fully vaccinated with an mRNA vaccine (Pfizer-BioNTech or Moderna) are less likely than unvaccinated persons to acquire SARS-CoV-2 or to transmit it to others. However, the risk for SARS-CoV-2 breakthrough infection in fully vaccinated people cannot be completely eliminated as long as there is continued community transmission of the virus.
- Studies are underway to learn more about the effectiveness of Johnson & Johnson/Janssen vaccine.
- At this time, there are limited data on vaccine effectiveness in people who are immunocompromised. People with immunocompromising conditions, including those taking immunosuppressive medications, should discuss the need for personal protective measures after vaccination with their healthcare provider.

All authorized COVID-19 vaccines demonstrated efficacy (range 65% to 95%) against symptomatic, laboratoryconfirmed COVID-19 in adults ≥18 years.

- For each authorized COVID-19 vaccine, efficacy was demonstrated across different populations, including
 elderly and younger adults, in people with and without underlying health conditions, and in people
 representing different races and ethnicities.
- The Pfizer-BioNTech COVID-19 vaccine also demonstrated high efficacy against symptomatic, laboratoryconfirmed COVID-19 in adolescents aged 12-17 years.

All authorized COVID-19 vaccines demonstrated high efficacy (≥89%) against COVID-19 severe enough to require hospitalization.

All authorized COVID-19 vaccines demonstrated high efficacy against COVID-19-associated death.

 In the clinical trials, no participants who received a COVID-19 vaccine died from COVID-19; the Moderna and Johnson & Johnson/Janssen trials among adults ≥18 years each had COVID-19 deaths in the unvaccinated placebo arm.

Preliminary data from the clinical trials among adults ≥18 years old suggest COVID-19 vaccination protects against symptomatic infection and may also protect against asymptomatic infection.

- In the Moderna trial, among people who had received a first dose, the number of asymptomatic people who tested positive for SARS-CoV-2 at their second-dose appointment was approximately 67% lower among vaccinees than among placebo recipients (0.1% and 0.3%, respectively)
- Efficacy of Johnson & Johnson/Janssen COVID-19 vaccine against asymptomatic infection was 74% in a subset of trial participants.

No trials have compared efficacy between any of the authorized vaccines in the same study population at the same time, making comparisons of efficacy difficult.

- All Phase 3 trials differed by calendar time and geography.
- Vaccines were tested in settings with different background COVID-19 incidence and circulating variants.

As members of the healthcare community, Phoenix Home Care & Hospice recognizes the importance of protecting our healthcare workers and reassuring our clients and communities that we can continue to safely care for them in their homes. We are dedicated to slowing the spread of COVID-19 and recommend all employees, contractors and staff receive a

COVID-19 vaccine.



Answering Your Questions About the New COVID-19 Vaccines

Do clinical trial results show whether vaccines are effective?

Yes. Clinical trials provide data and information about how well a vaccine prevents an infectious disease and about how safe it is. The Food and Drug Administration (FDA) evaluates these data, along with information from the manufacturer, to assess the safety and effectiveness of a vaccine. FDA then decides whether to approve a vaccine or authorize it for emergency use in the United States.

After a vaccine is either approved or authorized for emergency use by FDA, more assessments are done before a vaccine is recommended for public use. The goal of these assessments is to understand more about the protection a vaccine provides under real-world conditions, outside of clinical trials.

After COVID-19 vaccines are approved or authorized for emergency use by FDA and recommended for public use, CDC will further assess their effectiveness. These realworld assessments will compare groups of people who do and don't get vaccinated and people who do and don't get COVID-19 to find out how well COVID-19 vaccines are working to protect people.

Why would the effectiveness of vaccines be different after the clinical trials?

Many factors can affect a vaccine's effectiveness in real-world situations. These factors can include things such as how a vaccine is transported and stored or even how patients are vaccinated. Vaccine effectiveness can also be affected by differences in the underlying medical conditions of people vaccinated as compared to those vaccinated in the clinical trials.

Assessments of vaccine effectiveness can also provide important information about how well a vaccine is working in groups of people who were not included or were not well represented in clinical trials.

How will experts evaluate the COVID-19 vaccines in real-world conditions?

Experts are working on many types of real-world studies to determine vaccine effectiveness, and each uses a different method:

- Case-control studies will include cases (people who have the virus that causes COVID-19) and controls (people who do not have the virus that causes COVID-19). People who agree to participate in a case-control study will provide information on whether they received a COVID-19 vaccine or not. Experts will look to see if the cases were less likely to have received the vaccine than controls, which would show that the vaccine is working.
- A test-negative design study will enroll people who are seeking medical care for symptoms that could be due to COVID-19. In this special type of case-control study, experts will compare the COVID-19 vaccination status of those who test positive (meaning they have COVID-19) to those who test negative (meaning they do not have COVID-19).





www.cdc.gov/coronavirus/vaccines

- Cohort studies will follow people who have and haven't had a COVID-19 vaccine for several months to see if getting vaccinated protects them from getting the disease. This can be done in real time (prospectively) or by looking back in time (retrospectively) using data that were already collected, such as information in participants' medical records.
- Screening method assessments look at vaccination status among a group of cases (for example, cases detected through ongoing COVID-19 surveillance) and compares those cases with vaccination coverage among the overall population where those cases come from (for example people from the same state). By comparing coverage between these two groups, researchers can get an early estimate of whether a vaccine is working as expected.
- Ecologic analysis assessments look at groups of people

 such as those in different geographic locations or at different times to find out how many were vaccinated and how many were diagnosed with COVID-19. These analyses may be hard to interpret because the number of COVID-19 illnesses has changed rapidly over time and in different places.

CDC will use several methods because they can all contribute different information about how the vaccine is working.

Will assessments determine if the vaccines protect people from severe COVID-19 illness?

Yes. Severe illness from COVID-19 is defined as needing care in a hospital or intensive care unit (ICU), needing to be on a ventilator, or dying due to COVID-19.

- Experts will assess how well COVID-19 vaccines protect people against severe illness using case-control studies among hospitalized patients.
- Experts also will use cohort studies of electronic health records to see if people hospitalized with COVID-19 received the vaccine or not.

Will assessments determine if the vaccines protect people against mild illness?

Yes. CDC will use case-control studies to assess how well COVID-19 vaccines protect people against less severe forms of COVID-19 – for example, people with COVID-19 who need to visit a doctor but don't need to be hospitalized.

Will assessments determine if the vaccines protect people who are ill with no symptoms at all?

Yes. Some people can be infected with or "carry" the virus that causes COVID-19, but they don't feel sick or have any symptoms. Experts call this asymptomatic infection. It is important to know whether COVID-19 vaccines can help lower the number of people who have asymptomatic

infection. People with asymptomatic infection can unknowingly spread the virus to others.

A special type of cohort study will find out how effective the vaccine is when people are asymptomatic. People who agree to participate will be tested for COVID-19 every week whether they have symptoms or not. Experts will then compare the proportion of people with infection who were vaccinated to the proportion of people with infection who were not vaccinated.



Who will be included in the real-world vaccine assessments?

CDC is working to make sure real-world vaccine assessments include diverse groups of people including the following:

Healthcare personnel and essential workers

Experts will rapidly assess vaccine effectiveness among healthcare personnel working in hospitals, long term care/ skilled nursing facilities, or nursing homes in selected sites across the United States. These assessments will show how well COVID-19 vaccines protect healthcare personnel from getting sick or having severe illness. Assessments among healthcare personnel and essential workers will also inform how well COVID-19 vaccines protect them against getting infected, regardless of whether they have symptoms or not.

Older adults and those living in nursing homes

The risk for severe illness from COVID-19 increases with age, so making sure these vaccines protect older adults is critical. People living in nursing homes and long-term care facilities are at especially high risk of getting COVID-19 and severe disease. The Food and Drug Administration (FDA) and the Centers for Medicare and Medicaid Services (CMS) will use CMS Medicare billing data to assess COVID-19 vaccine effectiveness among older adults, including those living in nursing homes and long-term care facilities. These data will include information about whether people received a COVID-19 vaccine, whether they got sick with COVID-19, and if they needed hospital care. This information will help inform how well the vaccine works in preventing COVID-19 and severe illness among older adults.

Experts will also use data from CDC and CMS to conduct a case-control assessment. Experts will identify older adults hospitalized for COVID-19 and older adults hospitalized for other reasons. They will then compare how many cases and controls received a COVID-19 vaccine to estimate vaccine effectiveness.

People with underlying medical conditions

To better understand how well COVID-19 vaccines protect people with underlying medical conditions who may be at increased risk for severe illness. Experts are working to make sure various real-world vaccine assessments will include adults with heart conditions, obesity, and diabetes. The real-world vaccine effectiveness assessments will also collect information about other underlying medical conditions. This information will be used to better understand how well COVID-19 vaccines protect people with underlying medical conditions.

People in racial and ethnic minority groups

Long-standing systemic health and social inequities have put many people from racial and ethnic minority groups at increased risk of getting sick and dying from COVID-19. CDC is working to ensure that real-world assessments of vaccine effectiveness include diverse populations, such as people from racial and ethnic minority groups disproportionately affected by COVID-19.

CDC also is working with the Indian Health Service (IHS), tribal nations, and other partners to ensure that these realworld assessments include American Indian and Alaska Native populations who have been disproportionately affected by COVID-19. This is important to ensure that COVID-19 vaccines can help achieve health equity, so everyone has a fair opportunity to be as healthy as possible.

These vaccines were produced so quickly. How do we know they are safe?

It is the U.S. vaccine safety system's job to make sure that all vaccines are as safe as possible. Safety has been a top priority while federal partners have worked to make COVID-19 vaccines available for use in the United States.

The new COVID-19 vaccines have been evaluated in tens of thousands of individuals, who volunteered to be vaccinated and to participate in clinical trials. The information from these clinical trials allowed the U.S. Food and Drug Administration (FDA) to determine the safety and effectiveness of the vaccines. These clinical trials were conducted according to rigorous standards set forth by FDA.

FDA has determined that the newly authorized COVID-19 vaccines meet its safety and effectiveness standards. Therefore, FDA has made these vaccines available for use in the United States under what is known as an Emergency Use Authorization.



Will CDC continue to watch for problems with these new vaccines?

Yes. Even though no safety issues arose during the clinical trials, CDC and other federal partners will continue to monitor the new vaccines for serious side effects (known as adverse events) using many vaccine safety monitoring systems.

This continued monitoring can pick up on side effects that may not have been seen in clinical trials. If an unexpected side effect with the new COVID-19 vaccines is seen, experts can quickly study it further to determine if it is a true safety concern. Monitoring vaccine safety is critical to help ensure that the benefits of the COVID-19 vaccines continue to outweigh the risks for people who are vaccinated.

The current vaccine safety system is strong and robust, with the capacity to monitor COVID-19 vaccine safety effectively. Existing data systems can rapidly detect if a vaccine has any possible safety problems. These systems are being scaled up to fully meet the needs of the nation. Additional systems and data sources are also being developed to further enhance safety monitoring capabilities.

New vaccine safety monitoring systems and information sources

The following systems and information sources add another layer of safety monitoring, giving CDC and FDA the ability to evaluate COVID-19 vaccine safety in real time and make sure COVID-19 vaccines are as safe as possible:

 CDC: V-SAFE — A new smartphone-based, aftervaccination health checker for people who receive COVID-19 vaccines. V-SAFE will use text messaging and web surveys from CDC to check in with vaccine recipients for health problems following COVID-19 vaccination. The system also will provide telephone follow up to anyone who reports medically significant (important) adverse events.

CDC: National Healthcare Safety

Network (NHSN) — An acute care and long-term care facility monitoring system with reporting to the Vaccine Adverse Event Reporting System or VAERS

 FDA: Other large insurer/payer databases — A system of administrative and claims-based data for surveillance and research

Existing Safety Monitoring Systems

The safety of vaccines is monitored all the time with multiple approaches. As people get vaccinated, CDC, FDA, and other federal partners will use the following existing, robust systems and data sources to conduct ongoing safety monitoring in the following groups:

General public

- CDC and FDA: Vaccine Adverse Event Reporting System (VAERS) — The national system that collects reports from healthcare professionals, vaccine manufacturers, and the public of adverse events that happen after vaccination; reports of adverse events that are unexpected, appear to happen more often than expected, or have unusual patterns are followed up with specific studies
- CDC: Vaccine Safety Datalink (VSD) A network of 9 integrated healthcare organizations across the United States that conducts active surveillance and research; the system is also used to help determine whether possible side effects identified using VAERS are actually related to vaccination
- CDC: Clinical Immunization Safety Assessment (CISA)
 Project A collaboration between CDC and 7 medical research centers to provide expert consultation on individual cases and conduct clinical research studies about vaccine safety
- FDA and the Centers for Medicare and Medicaid Services: Medicare data — A claims-based system for active surveillance and research
- FDA: Biologics Effectiveness and Safety System (BEST)
 A system of electronic health record, administrative, and claims-based data for active surveillance and research
- FDA: Sentinel Initiative A system of electronic health record, administrative, and claims-based data for active surveillance and research

Members of the military

- Department of Defense (DOD): DOD VAERS data Adverse event reporting to VAERS for the DOD populations
- DOD: Vaccine Adverse Event Clinical System (VAECS)
 A system for case tracking and evaluation of adverse events following immunization in DOD and DOD-affiliated populations
- DOD: DOD Electronic Health Record and Defense Medical Surveillance System — A system of electronic health record and administrative data for active surveillance and research

Veterans

- Department of Veterans Affairs (VA): VA Adverse Drug Event Reporting System (VA ADERS) — A national reporting system for adverse events following receipt of drugs and immunizations
- VA Electronic Health Record and Active Surveillance
 System A system of electronic health record and administrative data for active surveillance and research

Tribal nations

 Indian Health Service (IHS): IHS VAERS data — Spontaneous adverse event reporting to VAERS for populations served by IHS and Tribal facilities



How mRNA COVID-19 Vaccines Work

TAUTA TAUTA

1111

Understanding the virus that causes COVID-19.

Coronaviruses, like the one that causes COVID-19, are named for the crown-like spikes on their surface, called **spike proteins**. These **spike proteins** are ideal targets for vaccines.

What is mRNA?

Messenger RNA, or mRNA, is genetic material that tells your body how to make proteins.

What is in the vaccine?

The vaccine is made of mRNA wrapped in a coating that makes delivery easy and keeps the body from damaging it.

How does the vaccine work?

The mRNA in the vaccine teaches your cells how to make copies of the **spike protein**. If you are exposed to the real virus later, your body will recognize it and know how to fight it off. The vaccine **DOES NOT** contain **ANY** virus, so it cannot give you COVID-19. It cannot change your DNA in any way. When your body responds to the vaccine, it can sometimes cause a mild fever, headache, or chills. This is completely normal and a sign that the vaccine is working.

Antibody

After the mRNA delivers the instructions, your cells break it down and get rid of it.

GETTING VACCINATED?

For information about COVID-19 vaccine, visit: cdc.gov/coronavirus/vaccines

Pfizer BioNTech Vaccine Moderna Vaccine



How Viral Vector COVID-19 Vaccines Work

Understanding the virus that causes COVID-19.

Coronaviruses, like the one that causes COVID-19, are named for the crown-like spikes on their surface, called **spike proteins**. These **spike proteins** are ideal targets for vaccines.

What is a viral vector vaccine?

A viral vector vaccine uses a harmless version of a different virus, called a "vector," to deliver information to the body that helps it protect you.

How does the vaccine work?

The vaccine teaches your body how to make copies of the **spike proteins**. If you are exposed to the real virus later, your body will recognize it and know how to fight it off. The vaccine **DOES NOT** contain the virus that causes COVID-19 and cannot give you COVID-19. It also cannot make you sick from the virus that is used as the vector. It cannot change your DNA in any way.

11111

When your body responds to the vaccine, it can sometimes cause tiredness, headache, muscle pain, nausea, or mild fever. These are normal signs the vaccine is working.

Antibody

GETTING VACCINATED?

For information about COVID-19 vaccine, visit cdc.gov/coronavirus/vaccines

Johnson & Johnson Janssen Vaccine

